

# **Earth's Future**

### **RESEARCH ARTICLE**

10.1029/2021EF002396

### **Key Points:**

- Escarpments have formed at the boundary between native prairies and cultivated fields in the Midwestern U.S. due to agricultural erosion
- Surveys of the escarpment heights at 20 sites indicate that the erosion rate, averaged since the time farming began, is 1.9 mm year<sup>-1</sup>
- Historically averaged erosion rates exceed model predictions of presentday rates, in part because the models do not include tillage erosion

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

E. A. Thaler, thaler.evan@gmail.com

### Citation:

Thaler, E. A., Kwang, J. S., Quirk, B. J., Quarrier, C. L., & Larsen, I. J. (2022). Rates of historical anthropogenic soil erosion in the Midwestern United States. *Earth's Future*, *10*, e2021EF002396. https://doi.org/10.1029/2021EF002396

Received 26 AUG 2021 Accepted 31 JAN 2022

### **Author Contributions:**

Conceptualization: Evan A. Thaler, Jeffrey S. Kwang, Isaac J. Larsen Data curation: Evan A. Thaler Formal analysis: Evan A. Thaler, Jeffrey S. Kwang Funding acquisition: Isaac J. Larsen Investigation: Evan A. Thaler, Jeffrey S. Kwang, Brendon J. Quirk, Caroline L. Quarrier Methodology: Evan A. Thaler, Jeffrey S. Kwang, Isaac J. Larsen Supervision: Isaac J. Larsen Validation: Evan A. Thaler, Jeffrey S. Kwang

© 2022 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

## **Rates of Historical Anthropogenic Soil Erosion in the Midwestern United States**

Evan A. Thaler<sup>1,2</sup>, Jeffrey S. Kwang<sup>1</sup>, Brendon J. Quirk<sup>1,3</sup>, Caroline L. Quarrier<sup>1</sup>, and Isaac J. Larsen<sup>1</sup>

<sup>1</sup>Department of Geosciences, Morrill Science Center, University of Massachusetts, Amherst, MA, USA, <sup>2</sup>Now at Earth and Environmental Sciences Division, Los Alamos National Laboratory, Los Alamos, NM, USA, <sup>3</sup>Now at Department of Earth, Atmospheric, and Planetary Sciences, Purdue University, West Lafayette, IN, USA

Abstract Erosion degrades soils, reduces crop yields, and diminishes ecosystem services. The total amount of soil that has been eroded since the initiation of farming is unknown in most agricultural landscapes, which hinders assessment of soil erosion trends. In the Midwestern U.S., erosion has caused native prairie remnants to become perched above surrounding farmland, providing an opportunity to measure historical soil erosion rates. We use high-resolution topographic surveys conducted across erosional escarpments at the boundary between 20 prairies and adjacent agricultural fields and show the median reduction in soil thickness ranges from 0.04 to 0.69 m, corresponding to erosion rates of 0.2-4.3 mm year<sup>-1</sup> with a median value of 1.9 mm year<sup>-1</sup>. We used an association between the measured reduction in soil thickness and topographic curvature to predict regional soil erosion integrated since the beginning of farming to the present. We estimate a median historical erosion rate of  $1.8 \pm 1.2$  mm year<sup>-1</sup>, which is nearly double the rate considered tolerable by the U.S. Department of Agriculture (USDA). Current soil loss predictions from the USDA National Resources Inventory (NRI) and the Daily Erosion Project (DEP) are lower than our historically averaged erosion rate by a factor of 3 and 8, respectively. We suggest that the NRI and the DEP underpredict soil loss rates because they do not include tillage erosion, a process shown to be important throughout the Midwestern U.S. Our findings indicate that further implementation of conservation practices is needed to reduce the high centennial-averaged soil erosion rates that we measure to sustainable levels.

**Plain Language Summary** The Midwestern United States is one of the world's most productive agricultural regions. However, high rates of soil erosion caused by farming have caused native prairie remnants to become perched above surrounding cultivated fields. We conducted high-resolution topographic surveys of the height of erosional escarpments at the borders of 20 native prairies and fields and used land records to estimate erosion rates that span the time since the initiation of farming to the present day. Our results indicate that soil thickness on hilltops in the Midwest has declined at an average rate of nearly 2 mm year<sup>-1</sup> over the past 150 years. The historical erosion rates exceed predictions of present-day erosion rates from national soil erosion assessments and levels considered tolerable by the U.S. Department of Agriculture.

### 1. Introduction

Conventional agricultural practices have accelerated soil erosion rates, resulting in widespread soil degradation throughout the world's agricultural regions (Montgomery, 2007b). Soil degradation diminishes soil fertility by removing organic matter and nutrients (Pimentel, 2006), which, without countervailing practices such as fertilization and genetic crop enhancements, leads to reductions in crop yields (Lal, 2004; Tilman et al., 2002). Fertilizer use, however, does not fully restore the productivity of eroded soils (Fenton et al., 2005), and because fossil fuels are required to generate the energy required for fertilizer production, the use of fertilizers to increase yields in degraded soils is not sustainable (Montgomery, 2007a). Further, soil erosion leads to increased agricultural production costs (Pimentel et al., 1995) and negative offsite effects such as increased sedimentation and nutrient export to downstream waterbodies (Tilman et al., 2002). In the United States, recognition of the high costs of soil erosion in the early twentieth century led to the development and implementation of soil conservation practices (Bennett, 1948). Field trials have demonstrated the efficacy of soil conservation efforts (Pimentel et al., 1976; Steiner, 1987), but it is unclear whether the advent of soil conservation practices and policies has led to a reduction of region-wide soil erosion rates in the U.S.





S. Kwang

Larsen

Visualization: Evan A. Thaler, Jeffrey

Writing – review & editing: Evan A. Thaler, Jeffrey S. Kwang, Brendon J.

Quirk, Caroline L. Quarrier, Isaac J.

Writing - original draft: Evan A. Thaler

Here, we measure the reduction in soil thickness within agricultural fields that has been caused by conventional farming activities. Soil erosion processes cause the decrease in soil thickness, and we equate the rate that the soil thickness has declined to a soil erosion rate. The decline in soil thickness caused by erosion leads to local removal of soil on hillslopes. Soil eroded from the hilltops is redistributed across the landscape by soil transport and depositional processes (e.g., Pennock & Frick, 2001) and can ultimately leave fields via transport by water (e.g., Quine et al., 1994).

Although some sedimentation studies have assessed anthropogenic soil erosion over long timescales (e.g., Heathcote & Downing, 2012; Heathcote et al., 2013), most studies tend to integrate over timescales limited to a few decades or less (García-Ruiz et al., 2015; Montgomery, 2007b; Figure S1 in Supporting Information S1). Remote sensing (e.g., Thaler, Larsen, et al., 2021) and soil survey-based methods (Jelinski & Yoo, 2016) can provide information on the areal extent of degraded soil, but such methods cannot quantify the total thickness of soil that has been eroded since the initiation of farming or rates of historical soil erosion. Hence in major agricultural regions, historical erosion rates averaged over the period following the initiation of farming to the present day are unknown. The lack of robust estimates of long-term agricultural erosion rates impedes our ability to assess temporal trends in soil erosion and evaluate the degree which recent conservation efforts have reduced erosion rates below historical levels.

Rates of soil erosion have been estimated from sediment budget studies, where a mass balance approach is applied by assuming that the volume of sediment delivery is proportional to the volume of soil eroded upstream (De Vente et al., 2007). Agricultural soil erosion can increase river sediment loads and sediment accumulation rates. For example, sedimentation rates in North America have increased tenfold following the initiation of agriculture (Kemp et al., 2020). However, the ability to directly relate fluvial sediment yields to erosion rates at cultivated upland sites is complicated by internal basin dynamics, including sediment storage within fields, floodplains, water bodies, and internally drained basins, the latter which are common in postglacial landscapes with hummocky topography (Lai & Anders, 2018). In the Midwest, sediment budget studies have reached different conclusions regarding trends in agricultural soil erosion. In the Driftless Area of Wisconsin, late twentieth century decreases in alluvial sedimentation rates suggest agricultural erosion rates declined following implementation of conservation practices (Trimble, 1999). However, alluvial sedimentation rates in the Sangamon River valley in Illinois have been relatively constant since 1870, suggesting soil erosion rates may not have decreased in the decades following the initiation of soil conservation efforts (Grimley et al., 2017). Fluvial sediment yields in other watersheds in the region, such as the Minnesota River, primarily reflect stream bank or bluff erosion related to Pleistocene river dynamics and recent changes in discharge, and hence cannot be used to infer erosion rates on agricultural uplands (Belmont et al., 2011; Gran et al., 2009, 2013; Schottler et al., 2014). Further, sediment yield studies do not provide information on the total magnitude of soil that has been eroded from uplands since the initiation of agriculture, and such studies still often rely on model predictions to infer historical changes in agricultural erosion rates (e.g., Trimble & Lund, 1982).

The U.S. Department of Agriculture National Resources Inventory (USDA NRI; U.S. Department of Agriculture, 2018) provides a subdecadal assessment of soil erosion rates at specific sites throughout the U.S. based on climate, soil property, and land use data. Site characteristics are used as parameters for the Revised Universal Soil Loss Equation (RUSLE; Renard et al., 1997) and the Wind Erosion Equation (Woodruff & Siddoway, 1965), which provide empirical estimates of soil loss by sheetwash and rill and wind erosion, respectively. A separate approach to assessing region-wide erosion, the Daily Erosion Project (DEP; Cruse et al., 2006; Gelder et al., 2018), uses the physically based Water Erosion Prediction Project (WEPP) model (Laflen & Flanagan, 2013), which integrates topography, precipitation, soil property, and crop data to provide daily estimates of the mass of soil eroded from hillslopes in Iowa, Nebraska, Minnesota, Wisconsin, and Kansas at the scale of ~90 km<sup>2</sup> watersheds. However, RUSLE and WEPP do not incorporate tillage or gully erosion in their model predictions, which can be important drivers of soil transport (Govers et al., 1996; S. Li et al., 2008; Öttl et al., 2021; Papiernik et al., 2009; J. Poesen, 2018; J. W. Poesen et al., 1996; Thaler, Larsen, et al., 2021; Valentin et al., 2005; Van Oost et al., 2006). NRI estimates of soil loss have been generated every 5 years from 1982 to 2017 and temporal trends suggest that erosion rates have generally been decreasing since 1982 (U.S. Department of Agriculture, 2018). Models used to assess changes in erosion rates for periods before and after the implementation of soil conservation practices in the U.S. predict large reductions in erosion (e.g., Trimble & Lund, 1982). However, the lack of measurements of upland soil erosion rates that span long timescales makes it difficult to empirically assess whether contemporary

erosion rates are substantially lower than rates prior to the mid-twentieth century. Such measurements are required to determine whether erosion rates have declined in the period following the advent of soil conservation practices.

The magnitude and temporal trends in soil erosion in the U.S. inferred from sediment budget and soil loss modeling do not agree, which has led to the conclusion that contemporary rates of soil erosion are poorly constrained (Trimble & Crosson, 2000). Here, we take a different approach to assessing erosion and report decreases in soil thickness based on field measurements of the height of topographic escarpments generated by agricultural erosion (e.g., Kaiser, 1961; Papendick & Miller, 1977) at the boundary between native prairie remnants and agricultural fields at sites throughout the Midwestern U.S. We relate the measurements of the reduction in soil thickness to topographic curvature and use high-resolution topographic data to generate a new regional estimate of soil erosion rates in the Midwest that spans the time from the initiation of agriculture in the mid-1800s to the present day and compare our results against predictions from the NRI and DEP.

### 2. Study Area

The Midwest is an important agricultural region that produces most of the corn and soybeans in the U.S. (U.S. Department of Agriculture, 2017). Here, we define the Midwest as the region in the midcontinental U.S. that was primarily tallgrass prairie or savanna prior to cultivation, though other ecosystems such as hardwood forests were more common in the east, due to greater precipitation (Figures S2 and S3 in Supporting Information S1; Carnahan, 1977; Daly et al., 1997). We specifically define our study area as portions of the states of North Dakota, South Dakota, Minnesota, Nebraska, Kansas, Missouri, Illinois, Indiana, and the entire state of Iowa (Figure 1). The study area generally falls within the extent of Pleistocene glaciation, except for part of north-east Kansas and the Driftless Area along the Wisconsin-Iowa border. All areas we study have high-resolution, lidarbased topographic data. The tallgrass prairie ecosystem led to the development of organic-carbon-rich mollisol soils (Samson & Knopf, 1994). Due to the productive soils, the prairies were rapidly converted to agricultural fields within a few decades of the arrival of European-Americans (Smith, 1992, 1998). Today less than 0.1% of the prairie remains (Samson & Knopf, 1994). Although most of the land in the Midwest has been converted to agricultural fields, small native prairie remnants, which were spared from cultivation, are preserved throughout the landscape (Figure 1). Because the native prairies have not been anthropogenically altered, they preserve the presettlement topography; natural erosion in the prairies occurs at such low rates that the topography can be considered static over post-European settlement timescales. For example, erosion rates measured using meteoric <sup>10</sup>Be concentrations from a prairie and a nearby agricultural field in Minnesota indicate that the natural erosion rate of the prairie is 0.047 mm year<sup>-1</sup> compared to an agricultural erosion rate of 3.09 mm year<sup>-1</sup> in the field (Jelinski et al., 2019). Our field observations indicate that when prairie remnants are located upgradient from adjacent agricultural fields, an escarpment is commonly present at the prairie-field boundary because agriculturally accelerated erosion within the fields outpaces natural erosion in the prairies (Figure 2), and the development of such escarpments is predicted by numerical models that simulate soil transport by tillage (e.g., Follain et al., 2006; S. Li et al., 2009; Van Oost et al., 2000; Vieira & Dabney, 2009). Hence, we measured the height of the erosional escarpment to quantify the decline in soil thickness that has occurred at these points on the landscape from the time cultivation began (approximately 1850–1900 CE) up to the present day. We use the field measurements in combination with high-resolution topographic data and a mathematical model of lateral soil transport to make spatially explicit predictions of the historical decline in soil thickness for agricultural fields across the Midwest.

### 3. Methods

### 3.1. Topographic Surveys

We used a real-time kinematic global positioning system (RTK GPS) to conduct cm-scale accuracy topographic surveys across the prairie-field boundary at 20 sites throughout the Midwest (Figure 1). A minimum of 10 transects were surveyed at each site, and transect locations were selected to span the range of topographic curvature at each site. At each transect, the height of the erosional escarpment was determined by using linear regression to extrapolate the elevation of the prairie surface across the erosional escarpment and into the field (Figure 2). The elevation at the base of the escarpment was determined by fitting a linear regression to survey points in the field. The escarpment height, which is equivalent to the reduction in soil thickness caused by erosion, was calculated





**Figure 1.** Location of the individual study sites (squares) within the Midwestern U.S. A region-wide decrease in soil thickness was predicted for areas with lidar topographic data, defined by the extent of the gray hillshade map. The blue outline shows the bounds of the Daily Erosion Project (DEP) data used in our study. The numbers correspond to each field site: 1, Willis; 2, Stinson; 3, Hayden; 4, Hoffman; 5, Kalsow; 6, Munson; 7, Steinauer; 8, Voight Pauper; 9, Kurtz; 10, Weston; 11, Greenlee; 12, McKnight; 13, Fricke; 14, Blue Gentian; 15, Judson; 16, Newell; 17, Loda; 18, Dinesen; 19, Harker; 20, Sheppard.

as the difference between the two regression lines at the base of the erosional escarpment. Use of the regression relationships accounts for the slope of the ground surface, resulting in more accurate determination of the reduction in soil thickness relative to simply comparing the elevation at the edge of the prairie versus the edge of the field, which would overestimate the change in soil thickness. In a few cases, the prairie and field slope in opposite directions, and rather than use linear regressions, the height of the escarpment was used to determine the change in soil thickness. Examples of escarpments, transects, and the inferred reduction in soil thickness are shown in Figure 2. At a small number of transects, deposition was observed and measured along the prairie-field boundary.

We estimated a soil erosion rate (Figure 3) by dividing the change in soil thickness by the time since the initiation of cultivation. Several of the prairies are pioneer cemeteries, and at these sites the date of the earliest gravestone was assumed to coincide with the initiation of cultivation. For the noncemetery sites, we assumed that cultivation began when the ownership of the land parcel was transferred from U.S. government to private individuals, based on records from the U.S. General Land Office Records (Bureau of Land Management, 2021). Dates of land transfer and hence the assumed initiation of cultivation range from 1846 to 1902. At 16 of the 20 study sites, the fields are still cultivated for row crop agriculture. The fields at the Loda, Judson, and Steinauer sites have been removed from row crop production within the last decade, but we do not know the exact year for each site. Therefore, we assumed that cultivated ceased in 2019 when we surveyed the fields, which leads to a conservative estimate of the erosion rate because the integration time is longer. The cessation of cultivation at the McKnight site occurred





**Figure 2.** Examples of erosional escarpments and real-time kinematic global positioning system (RTK GPS) survey data analysis. (a) Photograph of an escarpment between the native prairie (right) and the adjacent agricultural field (left) at Stinson Prairie. (b) An example survey transect and calculated elevation offset. In this example, the offset and change in soil thickness is 64 cm. (c) Map of topographic curvature, where red pixels indicate convex topography, and blue values indicate concave topography. GPS transect lines are shown as black dots, and the black stars indicate the transect shown in (a) and (b). The boundary between the prairie and field is shown as a magenta dashed line. (d) Photograph of an escarpment between the native prairie (top) and the adjacent agricultural field (bottom) at Willis Prairie. (e) An example profile of a surveyed transect and estimated elevation offset. In this example, the offset and change in soil thickness is 77 cm. (f) Same as (d), but for the Willis site. In (b) and (e), the prairie survey points and estimated topographic gradient are shown as the black points and line, respectively; the escarpment is shown as magenta points, and the points in the field and estimated topographic gradient are shown in red.

earlier, in the late 1960s, leading to an integrated erosion period of  $\sim 100$  years. The measurements of the decline in soil thickness integrate over the time since cultivation began and allow us to assess an average agricultural erosion rate but do not permit detection of temporal changes in erosion rates during the postsettlement period.

### 3.2. Topographic Controls on Historical Erosion Rates

Lidar point cloud data from the U.S. Geological Survey (USGS) National Map (Gesch et al., 2009) were used to generate 0.5 m spatial resolution digital elevation models (DEMs) for each site. We used a high-pass Gaussian filter from the *Python* package *astropy* (Astropy Collaboration et al., 2018) to smooth the DEMs to 4 m resolution, which is the optimal spatial resolution to calculate curvature for fields in this region (Thaler, Larsen, et al., 2021). Because the Gaussian filter samples from a spatial averaging window, the smoothing has the potential to greatly change the elevation values near the excarpment between the prairie and the field. To avoid this, we masked out the field-prairie boundary before applying the filter. We then calculated topographic curvature for each smoothed DEM and extracted the curvature values for each transect location.





**Figure 3.** Boxplots of the decline in soil thickness and estimated erosion rates at each study site. (a) Boxplot of the measured decline in soil thickness (m) for each of transect at each of the 20 sites. The median decline in soil thickness for all sites (0.3 m) is shown as the red dotted line. (b) Boxplot of erosion rates determined for each site. The U.S. Department of Agriculture (USDA) soil loss tolerance (*T*) value is the same for all sites  $(1 \text{ mm year}^{-1})$  and is shown as the gray dashed horizontal line. The red dotted line indicates the median value from all sites  $(1.9 \text{ mm year}^{-1})$ . Fifteen of the 20 sites have median erosion rates greater than the *T* value. Erosion rates were converted to a mass flux (t ha<sup>-1</sup> year<sup>-1</sup>) by assuming a uniform bulk density of 1,200 kg m<sup>-3</sup>. The numbers for each site correspond to the numbers in Figure 1. (c) Boxplot with compilation of previously published soil erosion rates (Montgomery, 2007b). The median global erosion rate in agricultural landscapes is 1.54 mm year<sup>-1</sup>. For each plot, the box spans the interquartile range; the black line is the median, and the bottom and top whiskers represent 5th and 95th percentiles, respectively. Outliers are shown as open circles. The colors of the boxes in (a) and (b) match those in Figures 1 and 4.

We fit a linear regression between the measured change in soil thickness and topographic curvature (Figure 4a) using data from all 20 study sites to derive a relationship that we used to predict changes in soil thickness throughout cultivated lands in the Midwest. We were unable to determine relationships between the change in soil thickness and curvature for each field site individually because many of the sites have a very limited range of curvature values, often with values only between -0.001 and  $0 \text{ m}^{-1}$  (Figure 4a), which precluded regression analysis. We used the regression relationship to estimate region-wide changes in soil thickness using topographic curvature values from 390,407 km<sup>2</sup> of lidar-derived DEM data. We estimated the uncertainty of the region-wide prediction using a Monte Carlo simulation which accounted for uncertainties in the slope and intercept of our empirical relationship between the change in soil thickness and topographic curvature. We randomly selected slope and intercept values from the compiled field sites dataset of soil thickness and topographic curvature 1,000 times and used the resulting 1,000 regression equations to estimate the change in soil thickness using curvature values from each cell in the full curvature dataset. The slope values were drawn from a normal distribution with the standard deviation (S.D.) defined by the 68th percentile confidence interval from our empirical fits (Figure 4a).





**Figure 4.** (a) Measurements of the reduction in soil thickness versus topographic curvature for each of the 304 transects at the 20 sites. The points are colored by site location, which matches colors in Figures 1 and 3. The gray shaded area represents the 68% prediction interval, which is  $\pm 0.22$  m. The equation of the regression line (mean  $\pm 68\%$  interval) for the full data set is  $y = (-28.9 \pm 6.3)x + (0.24 \pm 0.22)$ . (b) Erosion rate versus topographic curvature for each of the transects. The gray shaded area represents the 68% prediction interval, which is  $\pm 0.0015$  m year<sup>-1</sup>. The equation of the regression line for the full data set is  $y = (0.19 \pm 0.04)x + (0.002 \pm 0.0015)$ . The diffusion coefficient (*D*) calculated from the full data set is equivalent to the slope of the regression line,  $0.19 \pm 0.04$  m<sup>2</sup> year<sup>-1</sup>.

The *y*-intercept values were sampled from a normal distribution with the S.D. defined by the 68th percentile prediction intervals defined by the empirical measurements (Figure 4a).

### 3.3. Calibration of a Topographic Diffusion Coefficient

Our field measurements of the reduction in soil thickness integrate the combined influence of all erosion processes, but the greater reduction in soil thickness and erosion rates on more convex topography indicates a diffusive soil transport mechanism is the primary driver of erosion. Tillage erosion is the movement of soil by repeated tillage operations (Van Oost et al., 2006), where soil is preferentially removed from topographic convexities and deposited in topographic concavities, which causes landscapes to evolve via topographic diffusion. We used our estimates of the time-integrated soil erosion rate and measurements of topographic curvature to estimate a region-wide coefficient of topographic diffusion. The diffusion-like evolution of topography can be modeled as

$$dz/dt = D\nabla^2 z \tag{1}$$

where dz/dt is erosion rate averaged since cultivation began [L t<sup>-1</sup>], D is a diffusion coefficient  $[L^2 t^{-1}]$  that integrates factors that influence soil movement, such as tillage direction, tillage depth, and soil physical characteristics, and  $\nabla^2 z$  is the Laplacian operator of z, or topographic curvature [L<sup>-1</sup>], a measure of landscape convexity. By rearranging Equation 1, D can be estimated by calculating the slope of the regression line fit to the rate of soil thickness change calculated from the RTK GPS surveys (dz/dt) and land transfer records, and topographic curvature ( $\nabla^2 z$ ) data (Figure 4b). D and its  $\pm 1$  S.D. uncertainty were calculated from the slope of the regression line through the erosion rate and curvature data from all sites (Figure 4b). We also performed the same Monte Carlo analysis for regression between erosion rate and curvature values (Figure 4b) and assessed the uncertainty in our estimate of D using the 68% confidence interval of the slope value. The D value determined using this approach is a time-averaged value, as soil transport rates likely changed when non-mechanized agriculture was replaced by mechanized farming practices (Kwang et al., 2022).

Due to the low erosion rates in the prairies, there is virtually no downslope sediment flux from the prairies into the adjacent fields. The lack of upslope sediment supply to the field causes an additional elevation offset, as soil is moved downslope in the field but not replenished from upslope sources, a process which does not occur in areas without such a boundary. We evaluated the height of the escarpment that would be generated by such a no-flux boundary using an analytical model for hillslope evolution by diffusion described in Figures S4–11 in Supporting Information S1 and assess the influence of the no-flux boundary on our results.

### 3.4. Regional Estimates of the Historical Reduction in Soil Thickness

We calculated region-wide topographic curvature from lidar-derived topographic data at the same 4 m grid scale we used for the individual sites (Figure 1) and used the curvature data to empirically predict region-wide estimates of the reduction in soil thickness. The 4 m raster was first clipped to include only agricultural croplands using the Herbaceous Agriculture raster from the USGS Gap Analysis data set (Jennings, 2000). Waterways were removed from the DEM using a 50 m buffer from flowline centers identified by the National Hydrography

### **Earth's Future**





**Figure 5.** Example of predictions of the decline in soil thickness from  $16 \text{ km}^2$  of convex topography near Lakota, Iowa. Purple pixels indicate areas with lower reductions in soil thickness, while areas with a greater estimated reduction in soil thickness are shown green pixels. The total estimated mass of eroded soil in the full example extent is 98,784 metric tons. The inset demonstrates the removal of areas with buildings, roads, and fence lines from the analysis. Within the inset, we estimate 3,688 metric tons of eroded soil on 0.63 km<sup>2</sup> of convex topography.

Dataset (Horizon Systems Corporation, 2007), transit lines (roads, railways, etc.) were removed using a 30 m buffer from transit line centers within the National Transportation Dataset (U.S. Geological Survey, 2014), and property boundaries, which contain fence rows that can lead to anomalous curvature values, were removed using the Agricultural Conservation Planning Framework data set (Tomer et al., 2017). The buffer sizes were chosen to remove stream banks, ditches, and other infrastructure along transit lines that have curvature values which are not representative of cultivated fields.

For each pixel in the curvature grid, we predicted the total change in soil thickness and its uncertainty since the initiation of cultivation. The change in soil thickness was converted to a mass of eroded soil using an assumed uniform soil bulk density of 1,200 kg m<sup>-3</sup>, which is the median bulk density value for samples collected by the USDA Rapid Carbon Assessment within our study area (Wills et al., 2014). Because our surveys indicate the decline in soil thickness is the greatest on convex hillslopes, we only made predictions for the areas of fields with convex topography ( $\nabla^2 z < 0$ ), resulting in an area of 165,033 km<sup>2</sup> where we calculated the total mass of eroded soil. The total mass of soil removed from convex hillslopes was converted to an erosion rate by dividing by the mean time of cultivation for all of our study sites, which is 155 years. An example raster showing predictions of the decline in soil thickness for an area in north-central Iowa is shown in Figure 5.

### 3.5. Comparison Against Other Erosion Rate Estimates

We determined the annual soil loss (mm year<sup>-1</sup>) via water erosion estimated by the 2017 NRI soil erosion assessment (U.S. Department of Agriculture, 2018). The NRI erosion estimates are derived using RUSLE (Renard

et al., 1997) at target cells, which contain a surveyed field site. Wind erosion predictions from the NRI were not available to us, but with the exception of the Red River valley region in Minnesota and North Dakota, water erosion is predicted to be a more dominant erosion mechanism than wind in our study area (U.S. Department of Agriculture, 2000). The NRI methodology interpolates soil loss estimates to 2.5 km<sup>2</sup> spatial resolution for areas within 80 km of target cells. We determined the average annual hillslope soil loss rate predicted by the DEP from 2007 to 2020 by summing the total mass of soil transported to the base of hillslopes for each year and dividing by the number of years of predictions (13 years). Estimates of soil loss from the NRI and DEP have units of annual mass flux [M L<sup>-2</sup> t<sup>-1</sup>], and these values were converted to an erosion rate [L t<sup>-1</sup>] by assuming a uniform bulk density of 1,200 kg m<sup>-3</sup>. We then generated histograms and used a kernel density estimator to plot the probability distribution of the erosion rates derived from the NRI, the DEP, the rates calculated from each survey transect, and the rates estimated from our regional scaling. In addition to estimating erosion rates from the full extent of our regional analysis, the NRI and our regional scaling results were clipped to the extent of the DEP data, which has the smallest spatial extent of the three data sets, so that the same areas are compared.

To contextualize the soil erosion rates calculated from our survey data and the region-wide predictions, we compiled previously published soil erosion rates for croplands within the Midwestern U.S. We further place the erosion rates we measure within a global context by comparing erosion rates for each field site against a compilation of global agricultural erosion rate data (Montgomery, 2007b).

### 4. Results

The median reduction in soil thickness at the 20 field sites ranges from 0.004 to 0.69 m (Figure 3a). The median time-integrated erosion rates calculated from the changes in soil thickness range from 0.002 to 4.3 mm year<sup>-1</sup>, with a median value of 1.9 mm year<sup>-1</sup> for results across all 20 sites (Figure 3b). The greatest magnitudes of soil thickness change, and hence the greatest erosion rates, were measured in areas with the most convex topography ( $\nabla^2 z < 0$ ; Figure 4a). At eight transects in topographic concavities, the elevation of the field is higher than the elevation of the prairie due to soil deposition, and these transects are indicated by negative changes in soil thickness and erosion rate values (Figures 4a and 4b). The historical change in soil thickness measured from the escarpment heights is linearly correlated with topographic curvature, with a coefficient of determination ( $R^2$ ) of 0.21 and mean  $\pm 1$  S.D. slope and intercept values of  $-28.9 \pm 6.3$  and  $0.24 \pm 0.22$  m, respectively. Similarly, there is a linear relationship between erosion rate and topographic curvature (Figure 4b), with a predicted *D* value of  $0.19 \pm 0.04$  m<sup>2</sup> year<sup>-1</sup>.

Since the initiation of farming, we estimate the total mass of soil eroded from convex topography to be  $57.6 \times 10^9 \pm 37.8 \times 10^9$  metric tons across our study area. For the 165,033 km<sup>2</sup> of convex topography, the median historical averaged erosion rate equates to  $22.5 \pm 7.2$  metric tons ha<sup>-1</sup> year<sup>-1</sup> ( $1.8 \pm 1.2$  mm year<sup>-1</sup>). For the area of convex topography within the extent of the DEP data ( $105,519 \text{ km}^2$ ), our scaling analysis predicts a total of  $36.8 \times 10^9 \pm 13.2 \times 10^9$  metric tons of soil has been eroded, equating to a median erosion rate of  $14.3 \pm 9.2$  metric tons ha<sup>-1</sup> year<sup>-1</sup> ( $1.2 \pm 0.8$  mm year<sup>-1</sup>; Figure 6). The median rate of soil loss estimated from the NRI assessment, within the full extent of our study area is  $7.2 \pm 4.8$  metric tons ha<sup>-1</sup> year<sup>-1</sup> ( $0.6 \pm 0.4$  mm year<sup>-1</sup>), and the median rate estimated by the NRI within the extent of the DEP predictions is  $6.0 \pm 4.8$  metric tons ha<sup>-1</sup> year<sup>-1</sup> ( $0.5 \pm 0.4$  mm year<sup>-1</sup>; Figure 6). The median rate estimated by the DEP is  $2.4 \pm 6.0$  metric tons ha<sup>-1</sup> year<sup>-1</sup> ( $0.2 \pm 0.5$  mm year<sup>-1</sup>; Figure 6).

### 5. Discussion

### 5.1. Soil Erosion Rates in the Midwestern U.S.

Conversion of land from native tallgrass prairie to cultivation-based agriculture has caused soil thickness to decline by decimeters in the Midwestern U.S. The historical erosion rates documented by our surveys average over about 150 years and are consistent with previously measured soil erosion rates in the Midwest, which range from 0.14 to 7.7 mm year<sup>-1</sup> (Figure 3c). However, the previously documented erosion rates typically are based on only a few decades of measurement during the mid-twentieth century (Figure S1 in Supporting Information S1). Although our results cannot be used to assess whether soil erosion rates have increased or decreased over the last





**Figure 6.** (a) Histograms indicating the distribution of erosion rates within the extent of the Daily Erosion Project (DEP) analysis (Figure 1) estimated by the DEP, indicated by the solid red line, the USDA National Resources Inventory (NRI), shown as a blue line, erosion rates from previous studies in the Midwest are shown as a black line, rates calculated from the individual 304 surveyed transects collected from the 20 field locations in this study are shown as the teal line, and regional scale historical erosion rate estimates calculated in this study are shown as the brown line. (b) Kernel density estimates of values shown in (a). Estimates of regional historical erosion rates and NRI estimates within the full extent of our study area are shown in Figures S16 and S17 in Supporting Information S1.

150 years, the similarity between our results and the twentieth century, decadal-averaged erosion rates (Figure 3c) suggest that average erosion rates in the Midwestern landscape have been high for the entire period spanning the initiation of farming to the present day.

To set a goal for reduction of soil degradation, the USDA has assigned a soil loss tolerance (T) value to all mapped soil units. The T value is defined as the "maximum rate of annual soil loss that will permit crop productivity to be sustained economically and indefinitely" (U.S. Department of Agriculture, 2018). T values are based on multiple characteristics including soil physical properties and assumptions regarding soil formation rates (Skidmore, 1982). In the U.S., T values range from 2.2 to 11.2 metric tons ha<sup>-1</sup> year<sup>-1</sup> (U.S. Department of Agriculture, 2018), which is equivalent to 0.4–1 mm year<sup>-1</sup>, assuming a soil bulk density of 1,200 kg m<sup>-3</sup>. The T value for the soils at all our study sites is equivalent to 1 mm year<sup>-1</sup> (Figure 3), and it has been argued that such values are too high to adequately reduce soil loss to a sustainable level by balancing soil erosion and formation (Johnson, 1987; L. Li et al., 2009; Montgomery, 2007b). Our findings support those arguments; the median historical erosion rate exceeds the T value at 15 of our 20 field sites, indicating that soil erosion at these sites has outpaced even the most erosion-permissive soil conservation target set by the USDA for more than a century.

The five sites where our measured soil erosion rates are lower than T values are Kalsow, Loda, Dinesen, Fricke, and Sheppard. The Kalsow site has the second flattest topography in our data set, with a median topographic slope of 0.008 m m<sup>-1</sup> calculated from the survey data, which might explain the low erosion rates we measured there. However, the other four sites with erosion rates lower than the T value have median topographic slopes of 0.001, 0.03, 0.01, and 0.02 m m<sup>-1</sup>, respectively, which are greater than values at some sites where soil erosion rates outpace the T value, indicating that topography alone does not explain the relatively low erosion rates at the sites. Nor does soil parent material explain the lower erosion rates at some sites; the distribution of Quaternary deposits (Soller et al., 2009) indicates soils at the Dinesen and Sheppard sites formed from loess, whereas the soil parent material at the Kalsow, Harker, and Loda sites is glacial till. In the absence of clear topographic or parent material influences, it is possible that differences in historical agricultural practices, such as the frequency of tilling or the type of implements used for tilling, may contribute to differences in historical soil erosion rates across the sites. However, data on the history of farming practices at each site are not available to test that hypothesis.

### 5.2. Soil Redistribution Processes

The soil erosion we measure at each site is the sum of soil removal caused by all erosion processes, including rain splash, overland flow, biogenic soil creep, wind, and tillage. Our measurements cannot determine the relative importance of each mechanism. However, studies that have assessed the relative influence of tillage and water erosion demonstrate that both processes remove soil from convex topography, but that tillage tends to dominate, especially on the most convex parts of the landscape (e.g., Govers et al., 1996; Papiernik et al., 2009). The highest erosion rates we measured are on the most convex hilltops (Figure 4). Hence, we interpret the correspondence between topographic curvature and our measured erosion rates to indicate diffusive processes primarily reduces soil thickness on convex topography, particularly since convex ridgetop sites that we surveyed have little upslope accumulation area to generate overland flow but have lost decimeters of soil. Further, the *D* value calibrated from our measurements of erosion rate and topographic curvature ( $0.19 \pm 0.04 \text{ m}^2 \text{ year}^{-1}$ ; Figure 4b) is comparable to previously measured diffusion coefficients for tillage erosion, which range from 0.03 to 0.52 m<sup>2</sup> year<sup>-1</sup> (Van Oost et al., 2006), and indicates that tillage erosion plays a large role in redistributing soil from convex hillslopes to topographic convexities. These results provide further evidence that tillage erosion has been a primary driver of soil degradation in the Midwest (De Alba et al., 2004; S. Li et al., 2008; Thaler, Larsen, et al., 2021) and elsewhere (Montgomery, 2007a, 2007b).

We expect that the relative importance of water erosion increases downslope, as flow accumulation increases (e.g., Bryan, 2000; S. Li et al., 2007; Van Oost et al., 2005). Organic-carbon-rich A-horizon soils have been removed from a large proportion of agricultural fields in the Midwest, and about one third of the area with A-horizon removal has concave topography, suggesting that water erosion plays a significant role in removing soil from concave slopes (Thaler, Larsen, et al., 2021). We measured reductions in soil thickness for most of the survey transects in areas with concave topography, and water erosion may have been important in those locations.

The fate of soil eroded from convex topography cannot be assessed using the methodology presented in our study; however, both <sup>137</sup>Cs budget and modeling studies indicate that the combination of tillage and water erosion on convex slopes leads to redistribution of soil to concave landscape positions (Govers et al., 1996; Kwang et al., 2022; Papiernik et al., 2009; Quine et al., 1994; Van Oost et al., 2000, 2005). The soil that is deposited in topographic concavities can be exported to drainage networks, but the rates of deposition have been shown to be greater than rates of soil export via water erosion (Pennock & De Jong, 1987; Van Oost et al., 2000), leading to an overall increase in soil thickness in concavities within agricultural fields. Although we infer tillage is an important soil erosion processes, water and tillage erosion both degrade soils on convex topography. Given the longer transport distances for soil particles advected by water versus moved diffusively by tillage, water erosion is more likely to transport soil from fields and impact downstream water quality. Hence, conservation efforts designed to limit soil transport by both water and tillage erosion are valuable for reducing soil degradation and limiting offsite impacts to water resources.

### 5.3. Influence of Boundary Conditions and Topographic Diffusion on Erosion Rate Estimates

The height of the escarpment at the boundary between each field and prairie is influenced by two factors, the divergence of the soil flux in the direction parallel the boundary and the lack of soil flux across the boundary from the prairie to the field. The magnitude of boundary-parallel soil flux results in a reduction in the elevation of convex topography and deposition in areas with concave topography. However, in cases where the prairie slopes toward the field, the lack of soil flux perpendicular to the boundary results in a greater decline in soil thickness adjacent to the boundary than elsewhere within the field (Follain et al., 2006; Vieira & Dabney, 2009). If the reduction in soil thickness caused by the no-flux boundary is a large fraction of the total reduction in elevation or soil thickness, then our surveys may overestimate the decline in soil thickness and erosion rates relative to other locations within each field that have the same topography. We developed an analytical landscape evolution model that simulates topographic evolution via topographic diffusion, described in the Supporting Information S1, to separately quantify the contributions of boundary-parallel and boundary-perpendicular soil flux to the change in elevation that occurs near the prairie-field boundary.

The analytical model predicts that the change in elevation caused by the lack of soil flux across the boundary is a function of the initial, preagricultural hillslope angle. We use survey points in the prairies to estimate the initial slope at each transect and use the slope values to predict the magnitude of the elevation change caused by the

no-flux boundary using a 1-D analytical model (Figure S12 in Supporting Information S1). We assume that the slope of the prairie is representative of the preagricultural slope in the adjacent field.

For hillslopes with relief and slope lengths representative of our field sites and the median prairie slope value  $(0.04 \text{ m m}^{-1})$ , our 1-D analytical model predicts that the escarpment height caused by the no-flux boundary at individual cross sections is 25.5 cm, which is large relative to many of our measurements. However, the predicted elevation change caused by the no-flux boundary exceeds the surveyed elevation change at 38% of our surveyed cross sections (Figure S13 in Supporting Information S1). Hence, our 1-D analytical model overestimates the contribution of the no-flux boundary to total escarpment height; the values need to be reduced uniformly by 90% for all the predictions to be lower than the elevation changes we measured.

It is unclear why the analytical model overestimates the elevation change caused by the no-flux boundary, relative to our measurements. However, there are several lines of evidence that further suggest the no-flux boundary does not have a large influence on our erosion rate estimates. Tillage typically occurs in a direction that is parallel to the prairie-field boundary (Figure S14 in Supporting Information S1). Tracer studies have shown that tillage primarily moves soil in the same direction as tillage, such that soil flux in the direction perpendicular to tillage, or perpendicular to the prairie-field boundary, is 25% lower than the boundary-parallel soil flux (De Alba et al., 2001; Lindstrom et al., 1990). The anisotropy in soil flux caused by tillage is not accounted for in our model but doing so would presumably reduce the predicted elevation changes by 25%.

A value of *D* has been independently determined for three of the field sites, Stinson, Hoffman, and Willis using decreases in soil thickness inferred from satellite imagery and the depth dependence of soil spectral reflectance in the adjacent prairies (Kwang et al., 2022). The mean *D* value for the three sites of 0.25 m<sup>2</sup> year<sup>-1</sup> is of similar magnitude, though slightly higher than the value we have inferred from the topographic surveys and curvature data. The *D* values estimated by Kwang et al. (2022) are based on pixels throughout each field and hence are not influenced by the no-flux boundary. The similarity in *D* values indicates that the erosion rates we measure near the boundary are of similar magnitude to erosion rates that are occurring throughout each field, and hence that our survey-based erosion rates are not overestimates. The erosion rates we measure are also of similar order to prior measurements in the Midwest, though the median value we measure is about one third lower than the median from our compilation of prior measurements (Figure 3), which, again, suggests that the erosion rates we measure are not being greatly increased by the no-flux boundary.

Finally, if the no-flux boundary generates additional reductions in soil thickness along the boundary, then soils should be the most degraded at the boundary and become less degraded at increasing distance from the boundary. However, analysis of soil spectral properties related to the degree of soil degradation (Thaler et al., 2019) from soil topographic transects at Stinson, Hoffman, and Willis indicates that there is no pattern of increased soil degradation toward the boundary (Figure S15 in Supporting Information S1).

The topography of landscapes affected by diffusive soil transport processes becomes smoother over time. Landscape evolution modeling based on idealized sinusoidal topography suggests that landscape convexity has decreased by  $\sim 7\%$  since cultivation began, assuming hillslope lengths and wavelengths that are representative of our field sites. Since our calculation of *D* relies on the relationship between erosion rate and present-day topographic curvature, the predicted change in convexity suggests that our *D* value is a  $\sim 4\%$  overestimation (Supporting Information S1). The uncertainty in *D* caused by topographic change is small relative to the likely magnitude *D* has changed due to the conversion of prairies to farms. The *D* value we calculate is still 1–3 orders of magnitude greater than values measured in nonagricultural settings (Fernandes & Dietrich, 1997), which implies that agriculture has increased soil transport and erosion rates in the Midwest by orders of magnitude.

### 5.4. Comparison of Region-Wide Erosion Predictions

Soil loss predictions from RUSLE and WEPP are for the part of the landscape between where runoff initiates and where erosion by concentrated flow dominates (Gelder et al., 2018), whereas our predictions are for the part of the landscape with convex topography. The different model frameworks complicate direct comparison of model results, but all three approaches focus on upland parts of the landscape and exclude depositional areas. Inasmuch as the values can be directly compared, the median soil loss rate predicted by the NRI is a factor of 3 lower than our estimates, and the median soil loss rate predicted by the DEP is a factor of 8 lower. The lower rates predicted by the NRI and the DEP relative to the rates estimated by our analysis might suggest that modern erosion rates

are lower than historical values. Indeed, within the Midwestern U.S., subdecadal scale estimates of soil erosion from the NRI indicate that soil erosion rates decreased by 34% from 1982 to 2015 (U.S. Department of Agriculture, 2018). However, sedimentation data from lakes in Iowa indicate that most of the sediment has accumulated since 1950, which is temporally associated with the onset of agricultural intensification, rather than the initial land conversion to agriculture (Heathcote & Downing, 2012; Heathcote et al., 2013). The sedimentation rate data have led to the conclusion that conservation programs have not reduced downstream sediment delivery (Heathcote et al., 2013). Our regional estimates of soil erosion are based on the sum of all erosion processes (water, wind, tillage, etc.) captured by our field measurements. Whereas the NRI or the DEP predictions may be valid for assessing erosion by water, neither model includes tillage in their predictions of erosion, even though tillage is an important soil transport process in the Midwest (Papiernik et al., 2009; Thaler, Larsen, et al., 2021). Hence, rather than interpreting the difference between the historically averaged soil erosion rates we measure and modeled contemporary soil loss rates as a decline in erosion over time, we suggest the NRI and DEP may be underpredicting the current rates and magnitudes of soil loss in the Midwest because they do not incorporate the effects of tillage. Therefore, there is a need to incorporate existing tillage erosion models (De Alba, 2001; Kirkby et al., 2008; S. Li et al., 2008; Van Oost et al., 2005) within the models used to predict soil loss in the U.S. The NRI framework also currently does not include the ability to spatially resolve all relevant soil transport processes that lead to soil loss from convex hilltops, deposition in topographic lows, and transport of soil from fields, which hinders the ability to accurately predict contemporary rates of soil erosion, redistribution, and export from fields in the U.S.

### 6. Summary and Implications

When compared against *T* values and available long-term rates, our findings indicate that soil erosion rates in the Midwest are occurring at unsustainable levels and there is not a clear indication that the rates have declined since soil conservation practices and policies were implemented in the wake of the Dust Bowl of the 1930s. The erosion rates we estimate in the Midwest are consistent with rates in other agricultural regions across the globe  $(1.54 \pm 0.3 \text{ mm year}^{-1}; \text{ Montgomery, 2007b}; \text{ Figure 3d})$ , indicating that high rates of soil erosion are not unique to our study region. These similarities suggest that efforts must be taken at a global scale to reduce agriculturally induced soil erosion (Montgomery, 2017; Handelsman, 2021). Although there are political, social, and economic barriers to their implementation (Amundson & Biardeau, 2018; Schlesinger & Amundson, 2019), methods such as no-till farming (Montgomery, 2007b) and soil regenerative practices (Loisel et al., 2019; Montgomery, 2017) have the ability to reduce erosion rates. Incentivizing such practices (e.g., National Science and Technology Council, 2016) will likely be required to reduce soil erosion rates in the Midwest to levels that can sustain soil productivity, ecosystem services, and long-term prosperity (Amundson et al., 2015).

### **Data Availability Statement**

We have archived the topographic survey data from each of the 20 sites, shapefiles indicating the boundary of the prairie and field sites, digital elevation models for each of the sites, rasters containing values for the mean, minimum, and maximum volume of soil lost for each county in the study area, and the compilation of published soil erosion rates measured within the study area. The data are available at https://doi.org/10.7275/76w5-1z60 (Thaler, Kwang, et al., 2021).

### References

Amundson, R., Berhe, A. A., Hopmans, J. W., Olson, C., Sztein, A. E., & Sparks, D. L. (2015). Soil and human security in the 21st century. Science, 348(6235), 1261071.

Amundson, R., & Biardeau, L. (2018). Soil carbon sequestration is an elusive climate mitigation tool. Proceedings of the National Academy of Sciences of the United States of America, 115(46), 11652–11656. https://doi.org/10.1073/pnas.1815901115

Astropy Collaboration, Price-Whelan, A. M., Sipőcz, B. M., Günther, H. M., Lim, P. L., Crawford, S. M., et al. (2018). The Astropy Project: Building an open-science project and status of the v2.0 core package. *The Astronomical Journal*, *156*, 123. https://doi.org/10.3847/1538-3881/ aabc4f

Belmont, P., Gran, K. B., Schottler, S. P., Wilcock, P. R., Day, S. S., Jennings, C., et al. (2011). Large shift in source of fine sediment in the Upper Mississippi River. Environmental Science & Technology, 45(20), 8804–8810. https://doi.org/10.1021/es2019109

Bennett, H. H. (1948). Soil conservation: Hearings before the committee on agriculture, House of Representatives, Eightieth Congress, Second Session, March 1-5. US Government Printing Office.

#### Acknowledgments We thank Joe McGovern, Ryan Schmidt,

and Erin Van Waus of the Iowa Natural Heritage Foundation, John Pearson, Jon Judson, Lynn Sheppard, Paul Willis, Cyril and Darlene Thilges, Gerry Steinauer, David and Ruth Fritchner, the Cowling Arboretum at Carleton College, Jennifer Delisle, Melody Kroll, Rachel Simpson, Kristen Veum, John Peters, the Buena Vista County Conservation Board, the Kossuth County Conservation Board. Todd and Jane Gruis, James Ellis, Jeffrey Walk, Stephen Chaplin, and Alisha Paplow for assistance locating and accessing native prairie remnants. We also thank MyWay RTK and DigiFarm for providing accesses to RTK networks. Finally, we thank Tom Dunne and three anonymous reviewers whose comments greatly improved the manuscript. The research was funded by a Geological Society of America (GSA) Graduate Research grant and the Peter Birkeland Soil Geomorphology Award from the GSA Ouaternary Geology and Geomorphology Division to E.A.T. and NSF grant 1653191 to I.J.L.

- Bryan, R. B. (2000). Soil erodibility and processes of water erosion on hillslope. *Geomorphology*, 32(3–4), 385–415. https://doi.org/10.1016/ s0169-555x(99)00105-1
- Bureau of Land Management. (2021). "Land Patent Search." Database and images. https://glorecords.blm.gov/
- Carnahan, J. A. (1977). Potential Natural Vegetation of the Conterminous United States, A.W. Kuchler, Special Publication No. 36. American Geographical Society. New York. 1975. Second edition. 995 × 1545 mm; folded map sheet, 1: 3 168 000 map. *Cartography*, *10*, 55.
- Cruse, R., Flanagan, D., Frankenberger, J., Gelder, B., Herzmann, D., James, D., et al. (2006). Daily estimates of rainfall, water runoff, and soil erosion in Iowa. *Journal of Soil and Water Conservation*, 61(4), 191–199.
- Daly, C., Taylor, G., & Gibson, W. (1997). The PRISM approach to mapping precipitation and temperature. In Proceedings of the 10th AMS Conference on Applied Climatology (pp. 20–23).
- De Alba, S. (2001). Modeling the effects of complex topography and patterns of tillage on soil translocation by tillage with mouldboard plough. *Journal of Soil and Water Conservation*, 56(4), 335–345.
- De Alba, S., Lindstrom, M., Schumacher, T. E., & Malo, D. D. (2004). Soil landscape evolution due to soil redistribution by tillage: A new conceptual model of soil catena evolution in agricultural landscapes. *Catena*, 58(1), 77–100. https://doi.org/10.1016/j.catena.2003.12.004
- De Vente, J., Poesen, J., Arabkhedri, M., & Verstraeten, G. (2007). The sediment delivery problem revisited. *Progress in Physical Geography*, 31(2), 155–178. https://doi.org/10.1177/0309133307076485
- Fenton, T. E., Kazemi, M., & Lauterbach-Barrett, M. A. (2005). Erosional impact on organic matter content and productivity of selected Iowa soils. Soil and Tillage Research, 81, 163–171. https://doi.org/10.1016/j.still.2004.09.005
- Fernandes, N. F., & Dietrich, W. E. (1997). Hillslope evolution by diffusive processes: The timescale for equilibrium adjustments. Water Resources Research, 33(6), 1307–1318. https://doi.org/10.1029/97WR00534
- Follain, S., Minasny, B., McBratney, A. B., & Walter, C. (2006). Simulation of soil thickness evolution in a complex agricultural landscape at fine spatial and temporal scales. *Geoderma*, 133(1–2), 71–86. https://doi.org/10.1016/j.geoderma.2006.03.038
- García-Ruiz, J. M., Beguería, S., Nadal-Romero, E., González-Hidalgo, J. C., Lana-Renault, N., & Sanjuán, Y. (2015). A meta-analysis of soil erosion rates across the world. *Geomorphology*, 239, 160–173. https://doi.org/10.1016/j.geomorph.2015.03.008
- Gelder, B., Sklenar, T., James, D., Herzmann, D., Cruse, R., Gesch, K., & Laflen, J. (2018). The Daily Erosion Project—Daily estimates of water runoff, soil detachment, and erosion. *Earth Surface Processes and Landforms*, 43(5), 1105–1117. https://doi.org/10.1002/esp.4286
- Gesch, D., Evans, G., Mauck, J., Hutchinson, J., & Carswell, W. J., Jr. (2009). The national map—Elevation. US Geological Survey Fact Sheet, 3053(4). https://www.usgs.gov/programs/national-geospatial-program/national-map
- Govers, G., Quine, T. A., Desmet, P. J., & Walling, D. E. (1996). The relative contribution of soil tillage and overland flow erosion to soil redistribution on agricultural land. *Earth Surface Processes and Landforms*, 21(10), 929–946. https://doi.org/10.1002/ (SICI)1096-9837(199610)21:10<929::AID-ESP631>3.0.CO;2-C
- Gran, K. B., Belmont, P., Day, S. S., Jennings, C., Johnson, A., Perg, L., et al. (2009). Geomorphic evolution of the Le Sueur River, Minnesota, USA, and implications for current sediment loading. In *Management and restoration of fluvial systems with broad historical changes and human impacts: Geological Society of America Special Paper* (Vol. 451, pp. 119–130). https://doi.org/10.1130/2009.2451(08)
- Gran, K. B., Finnegan, N., Johnson, A. L., Belmont, P., Wittkop, C., & Rittenour, T. (2013). Landscape evolution, valley excavation, and terrace development following abrupt postglacial base-level fall. *Geological Society of America Bulletin*, 125(11–12), 1851–1864. https://doi. org/10.1130/B30772.1
- Grimley, D. A., Anders, A. M., Bettis, E. A., Bates, B. L., Wang, J. J., Butler, S. K., et al. (2017). Using magnetic fly ash to identify post-settlement alluvium and its record of atmospheric pollution, central USA. *Anthropocene*, *17*, 84–98. https://doi.org/10.1016/j.ancene.2017.02.001 Handelsman, J. (2021). A world without soil: The past, present, and precarious future of the earth beneath our feet. Yale University Press.
- Heathcote, A. J., & Downing, J. A. (2012). Impacts of eutrophication on carbon burial in freshwater lakes in an intensively agricultural landscape. *Ecosystems*, 15(1), 60–70. https://doi.org/10.1007/s10021-011-9488-9
- Heathcote, A. J., Filstrup, C. T., & Downing, J. A. (2013). Watershed sediment losses to lakes accelerating despite agricultural soil conservation efforts. *PLoS One*, 8(1), e53554. https://doi.org/10.1371/journal.pone.0053554
- Horizon Systems Corporation. (2007). National Hydrography Dataset Plus: Documentation. Retrieved from http://www.horizon-systems.com/ nhdplus/documentation.php
- Jelinski, N. A., Campforts, B., Willenbring, J. K., Schumacher, T. E., Li, S., Lobb, D. A., et al. (2019). Meteoric beryllium-10 as a tracer of erosion due to post settlement land use in west-central Minnesota, USA. Journal of Geophysical Research: Earth Surface, 124, 874–901. https://doi. org/10.1029/2018JF004720
- Jelinski, N. A., & Yoo, K. (2016). The distribution and genesis of eroded phase soils in the conterminous United States. *Geoderma*, 279, 149–164. Jennings, M. D. (2000). Gap Analysis: Concepts, methods, and recent results. *Landscape Ecology*, 15(1), 5–20. https://doi.org/10.1023/a:1008184408300
- Johnson, L. C. (1987). Soil loss tolerance: Fact or myth? Journal of Soil Water Conservation, 42, 155-160.
- Kaiser, V. G. (1961). Historical land use and erosion in the Palouse-A reappraisal. Northwest Science, 35, 139-153.
- Kemp, D. B., Sadler, P. M., & Vanacker, V. (2020). The human impact on North American erosion, sediment transfer, and storage in a geologic context. *Nature Communications*, 11, 1–9. https://doi.org/10.1007/978-3-476-05728-0\_3508-1
- Kirkby, M., Irvine, B., Jones, R. J., Govers, G., & team, P. (2008). The PESERA coarse scale erosion model for Europe. I.—Model rationale and implementation. *European Journal of Soil Science*, 59(6), 1293–1306. https://doi.org/10.1111/j.1365-2389.2008.01072.x
- Kwang, J. S., Thaler, E. A., Quirk, B. J., Quarrier, C. L., & Larsen, I. J. (2022). A landscape evolution modeling approach for predicting three-dimensional soil organic carbon redistribution in agricultural landscapes. *Journal of Geophysical Research: Biogeosciences*. https://doi. org/10.1029/2021JG006616
- Laflen, J. M., & Flanagan, D. C. (2013). The development of U. S. soil erosion prediction and modeling. International Soil and Water Conservation Research, 1(2), 1–11. https://doi.org/10.1016/s2095-6339(15)30034-4
- Lai, J., & Anders, A. M. (2018). Modeled postglacial landscape evolution at the southern margin of the Laurentide ice sheet: Hydrological connection of uplands controls the pace and style of fluvial network expansion. *Journal of Geophysical Research: Earth Surface*, 123, 967–984. https://doi.org/10.1029/2017JF004509
- Lal, R. (2004). Soil carbon sequestration impacts on global climate change and food security. Science, 304(5677), 1623–1627. https://doi.org/10.1126/science.1097396
- Li, L., Du, S., Wu, L., & Liu, G. (2009). An overview of soil loss tolerance. *Catena*, 78(2), 93–99. https://doi.org/10.1016/j.catena.2009.03.007 Li, S., Lobb, D. A., Lindstrom, M. J., & Farenhorst, A. (2007). Tillage and water erosion on different landscapes in the northern North American
- Great Plains evaluated using <sup>137</sup>Cs technique and soil erosion models. *Catena*, 70(3), 493–505. https://doi.org/10.1016/j.catena.2006.12.003 Li, S., Lobb, D. A., Lindstrom, M. J., Papiernik, S. K., & Farenhorst, A. (2008). Modeling tillage-induced redistribution of soil mass and its constituents within different landscapes. *Soil Science Society of America Journal*, 72(1), 167–179. https://doi.org/10.2136/sssaj2006.0418



- Li, S., Lobb, D. A., & Tiessen, K. H. (2009). Modeling tillage-induced morphological features in cultivated landscapes. *Soil and Tillage Research*, 103(1), 33–45. https://doi.org/10.1016/j.still.2008.09.005
- Lindstrom, M. J., Nelson, W. W., Schumacher, T. E., & Lemme, G. D. (1990). Soil movement by tillage as affected by slope. Soil and Tillage Research, 17(3–4), 255–264. https://doi.org/10.1016/0167-1987(90)90040-k
- Loisel, J., Connors, J. P. C., Hugelius, G., Harden, J. W., & Morgan, C. L. (2019). Soils can help mitigate CO<sub>2</sub> emissions, despite the challenges. Proceedings of the National Academy of Sciences of the United States of America, 116(21), 10211–10212. https://doi.org/10.1073/pnas.1900444116

Montgomery, D. R. (2007a). Dirt: The erosion of civilizations. University of California Press.

- Montgomery, D. R. (2007b). Soil erosion and agricultural sustainability. Proceedings of the National Academy of Sciences of the United States of America, 104(33), 13268–13272. https://doi.org/10.1073/pnas.0611508104
- Montgomery, D. R. (2017). Growing a revolution: Bringing our soil back to life. W. W. Norton & Company.
- National Science and Technology Council. (2016). The state and future of U.S. soils. Retrieved from https://obamawhitehouse.archives.gov/sites/ default/files/microsites/ostp/ssiwg\_framework\_december\_2016.pdf

Öttl, L. K., Wilken, F., Auerswald, K., Sommer, M., Wehrhan, M., & Fiener, P. (2021). Tillage erosion as an important driver of in-field biomass patterns in an intensively used hummocky landscape. *Land Degradation & Development*, 32, 3077–3091.

Papendick, R. I., & Miller, D. E. (1977). Conservation tillage in the Pacific Northwest. Journal of Soil and Water Conservation, 32, 49–56.

- Papiernik, S., Schumacher, T., Lobb, D., Lindstrom, M., Lieser, M., Eynard, A., et al. (2009). Soil properties and productivity as affected by topsoil movement within an eroded landform. *Soil and Tillage Research*, 102(1), 67–77. https://doi.org/10.1016/j.still.2008.07.018
- Pennock, D. J., & De Jong, E. (1987). The influence of slope curvature on soil erosion and deposition in hummock terrain. Soil Science, 144(3), 209–217. https://doi.org/10.1097/00010694-198709000-00007
- Pennock, D. J., & Frick, A. H. (2001). The role of field studies in landscape-scale applications of process models: An example of soil redistribution and soil organic carbon modeling using CENTURY. Soil and Tillage Research, 58, 183–191. https://doi.org/10.1016/s0167-1987(00)00167-7
- Pimentel, D. (2006). Soil erosion: A food and environmental threat. Environment, Development and Sustainability, 8(1), 119–137. https://doi.org/10.1007/s10668-005-1262-8
- Pimentel, D., Harvey, C., Resosudarmo, P., Sinclair, K., Kurz, D., McNair, M., et al. (1995). Environmental and economic costs of soil erosion and conservation benefits. *Science*, 267, 1117–1123. https://doi.org/10.1126/science.267.5201.1117
- Pimentel, D., Terhune, E. C., Dyson-Hudson, R., Rochereau, S., Samis, R., Smith, E. A., et al. (1976). Land degradation: Effects on food and energy resources. *Science*, 194(4261), 149–155. https://doi.org/10.1126/science.194.4261.149
- Poesen, J. (2018). Soil erosion in the Anthropocene: Research needs. Earth Surface Processes and Landforms, 43(1), 64-84. https://doi. org/10.1002/esp.4250
- Poesen, J. W., Vandaele, K., & Van Wesemael, B. (1996). Contribution of gully erosion to sediment production on cultivated lands and rangelands. IAHS Publications-Series of Proceedings and Reports-Intern Assoc Hydrological Sciences, 236, 251–266.
- Quine, T. A., Desmet, P. J., Govers, G., Vandaele, K., & Walling, D. E. (1994). A comparison of the roles of tillage and water erosion in landform development and sediment export on agricultural land near Leuven, Belgium. *IAHS Publications-Series of Proceedings and Reports-Intern* Assoc Hydrological Sciences, 224, 77–86.
- Renard, K. G., Foster, G. R., Weesies, G. A., McCool, D. K., & Yoder, D. C. (1997). Predicting soil erosion by water: A guide to conservation planning with the Revised Universal Soil Loss Equation (RUSLE). USDA-ARS Agriculture Handbook 703.
- Samson, F., & Knopf, F. (1994). Prairie conservation in north America. BioScience, 44(6), 418-421. https://doi.org/10.2307/1312365
- Schlesinger, W. H., & Amundson, R. (2019). Managing for soil carbon sequestration: Let's get realistic. *Global Change Biology*, 25(2), 386–389. https://doi.org/10.1111/gcb.14478
- Schottler, S. P., Ulrich, J., Belmont, P., Moore, R., Lauer, J. W., Engstrom, D. R., et al. (2014). Twentieth century agricultural drainage creates more erosive rivers. *Hydrological Processes*, 28(4), 1951–1961. https://doi.org/10.1002/hyp.9738
- Skidmore, E. L. (1982). Soil loss tolerance. In B. L. Schmidt, R. R. Allmaras, J. V. Mannering & R. I. Papendick (Eds.), Determinants of Soil Loss Tolerance (Special Publication, Vol. 45, pp. 87–93). American Society of Agronomy.
- Smith, D. D. (1992). Tallgrass prairie settlement: Prelude to the demise of the tallgrass ecosystem. In Proceedings of the Twelfth North American Prairie Conference (pp. 195–199). University of Northern Iowa.
- Smith, D. D. (1998). Iowa prairie: Original extent and loss, preservation and recovery attempts. Journal of the Iowa Academy of Science, 105(3), 94–108.
- Soller, D. R., Reheis, M. C., Garrity, C. P., & Van Sistine, D. R. (2009). Map database for surficial materials in the Conterminous United States. US Geological Survey Data Series, 425.
- Steiner, F. (1987). Soil conservation policy in the United States. Environmental Management, 11(2), 209–223. https://doi.org/10.1007/bf01867200
  Thaler, E. A., Kwang, J. S., & Larsen, I. J. (2021). Historical rates of soil erosion in the Midwestern U.S. Data and Datasets, 138. https://doi.org/10.7275/76w5-1260
- Thaler, E. A., Larsen, I. J., & Yu, Q. (2019). A new index for remote sensing of soil organic carbon based solely on visible wavelengths. Soil Science Society of America Journal, 83(5), 1443–1450. https://doi.org/10.2136/sssaj2018.09.0318
- Thaler, E. A., Larsen, I. J., & Yu, Q. (2021). The extent of soil loss across the US Corn Belt. Proceedings of the National Academy of Sciences of the United States of America, 118(8), e1922375118. https://doi.org/10.1073/pnas.1922375118
- Tilman, D., Cassman, K. G., Matson, P. A., Naylor, R., & Polasky, S. (2002). Agricultural sustainability and intensive production practices. *Nature*, 418(6898), 671–677. https://doi.org/10.1038/nature01014
- Tomer, M. D., James, D. E., & Sandoval-Green, C. M. (2017). Agricultural conservation planning framework: 3. Land use and field boundary database development and structure. *Journal of Environmental Quality*, 46, 676–686. https://doi.org/10.2134/jeq2016.09.0363
- Trimble, S. W. (1999). Decreased rates of alluvial sediment storage in the Coon Creek Basin, Wisconsin, 1975-93. *Science*, 285(5431), 1244–1246. https://doi.org/10.1126/science.285.5431.1244
- Trimble, S. W., & Crosson, P. (2000). US soil erosion rates—Myth and reality. Science, 289(5477), 248–250. https://doi.org/10.1126/ science.289.5477.248
- Trimble, S. W., & Lund, S. W. (1982). Soil conservation and the reduction of erosion and sedimentation in the Coon Creek Basin, Wisconsin (Vol. 1234). US Government Printing Office.
- U.S. Department of Agriculture. (2000). Summary Report: 1997 National Resources Inventory (revised December 2000) (pp. 89). Natural Resources Conservation Service/Statistical Laboratory, Iowa State University.
- U.S. Department of Agriculture. (2017). "Crop production". Retrieved from https://www.nass.usda.gov/Publications/Todays\_Reports/reports/ crop1117.pdf

- U.S. Department of Agriculture. (2018). Summary Report: 2017 National Resources Inventory. Natural Resources Conservation Service/ Statistical Laboratory, Iowa State University.
- U.S. Geological Survey. (2014). USGS National Transportation Dataset. https://data.usgs.gov/datacatalog/data/ USGS:ad3d631d-f51f-4b6a-91a3-e617d6a58b4e
- Valentin, C., Poesen, J., & Li, Y. (2005). Gully erosion: Impacts, factors and control. Catena, 63(2-3), 132–153. https://doi.org/10.1016/j. catena.2005.06.001
- Van Oost, K., Govers, G., De Alba, S., & Quine, T. A. (2006). Tillage erosion: A review of controlling factors and implications for soil quality. Progress in Physical Geography: Earth and Environment, 30(4), 443–466. https://doi.org/10.1191/0309133306pp487ra
- Van Oost, K., Govers, G., & Desmet, P. (2000). Evaluating the effects of changes in landscape structure on soil erosion by water and tillage. Landscape Ecology, 15(6), 577–589. https://doi.org/10.1023/a:1008198215674
- Van Oost, K., Govers, G., Quine, T. A., Heckrath, G., Olesen, J. E., De Gryze, S., et al. (2005). Landscape-scale modeling of carbon cycling under the impact of soil redistribution: The role of tillage erosion. *Global Biogeochemical Cycles*, 19, GB4014. https://doi.org/10.1029/2005GB002471
   Vieira, D. A. N., & Dabney, S. M. (2009). Modeling landscape evolution due to tillage: Model development. *Transactions of the ASABE*, 52(5),
- 1505–1522. https://doi.org/10.13031/2013.29139
- Wills, S., Loecke, T., Sequeira, C., Teachman, G., Grunwald, S., & West, L. T. (2014). Overview of the US Rapid Carbon Assessment project: Sampling design, initial summary and uncertainty estimates. In Soil carbon (pp. 95–104). Springer. https://doi.org/10.1007/978-3-319-04084-4\_10
- Woodruff, N. P., & Siddoway, F. H. (1965). A wind erosion equation. Soil Science Society of America Journal, 29(5), 602–608. https://doi.org/10.2136/sssaj1965.03615995002900050035x

### **Reference From the Supporting Information**

Dabney, S. M., Liu, Z., Lane, M., Douglas, J., Zhu, J., & Flanagan, D. C. (1999). Landscape benching from tillage erosion between grass hedges. Soil and Tillage Research, 51(3-4), 219–231. https://doi.org/10.1016/s0167-1987(99)00039-2